

Potsdam-Institut für Klimafolgenforschung

Originally published as:

Schmid, E., Pahle, M., Knopf, B. (2013): Renewable electricity generation in Germany: A meta-analysis of mitigation scenarios. - Energy Policy, 61, 1151-1163

DOI: <u>10.1016/j.enpol.2013.06.105</u>

Renewable Electricity Generation in Germany: A Meta-Analysis of Mitigation Scenarios

Eva Schmid^{1a}, Michael Pahle^a, Brigitte Knopf^a

^aPotsdam Institute for Climate Impact Research (PIK), P.O. Box 601203, 14412 Potsdam, Germany

Abstract

German energy policy targets envision a share of electricity from renewable energy sources (RES-E) of at least 80% in 2050. How can the transformation of the German electricity sector be achieved and at what costs? This paper addresses these questions by means of a meta-analysis of ten recent model-based mitigation scenarios for Germany. It turns out that the scenarios exploit the three basic strategic options of increasing the share of RES-E—domestic RES-E generation, electricity demand reductions, and RES-E imports—to substantially different extents. Domestic RES-E generation increases in all scenarios, particularly from onshore and offshore wind. Scenarios that rely heavily on reducing electricity demand require a relatively low expansion of domestic RES-E generation. Despite detailed technical analyses, insights on the costs of the transformation remain limited. A discussion of underlying scenario assumptions reveals that it is unclear whether (i) RES-E and system integration technology development will be as cost-competitive as postulated, (ii) implicitly assumed institutional requirements will be realized, and (iii) relevant actors in the transformation process will be incentivized accordingly. Therefore, future research should pursue a thorough assessment of strategic options for transforming the German electricity system that consistently integrates technologies, institutions, and actors.

Keywords: energy strategy, energy system model, transformation costs

¹ Corresponding author, Tel.: +49 331 288 2674, Fax: +49 331 288 2570 *Email address:* eva.schmid@pik-potsdam.de (Eva Schmid)

1 Introduction

During the recent years German energy policy has undergone substantial changes, particularly so regarding the electricity sector. In June 2011, after the events in Fukushima, the German Government reconfirmed the accelerated phase-out of nuclear electricity generation until 2022, thus withdrawing the prolongation of the phase-out of on average 12 years that was passed only the year before. At the same time, it amended the target formulation of the Renewable Energy Sources Act, which now specifies an explicit pathway for the development of the share of electricity from renewable energy sources (RES-E) that reaches at least 80% by 2050, with interim steps of 35%, 50%, and 75% by 2020, 2030 and 2040, respectively. The so-called "Energiewende" (energy transition) is by now globally renowned as the first attempt of an industrialized country to decarbonize its energy system solely by means of renewables and energy efficiency improvements (Schiermeier, 2013), embracing to reduce CO₂ emissions by 80-95% in 2050 relative to 1990 (Federal Government, 2010).

But how can Germany attain the long-term target of at least 80% electricity provision from renewable energy sources by 2050, and what are the costs? Finding answers to these questions is clearly of major interest to the German policymaker, and also to the rest of the world observing this experiment from afar. Fundamentally, there are three strategic options to alter the share of RES-E: Increasing domestic RES-E generation, decreasing electricity demand and increasing RES-E imports (Figure 1). In recent years, particularly the option to increase domestic RES-E generation has been exploited. Since the introduction of the Renewable Energy Sources Act in 2000, domestic RES-E generation has continuously increased from 37 TWh to 136 TWh in 2012 (BMWi, 2013), resulting in a share of RES-E share of 22% in 2012 (BMWi, 2013). Hence, there is still a long way to go to attain the 80% target.



Figure 1. Conceptual overview of strategic options for the long-term transformation of the German electricity sector to attain a share of at least 80% renewable electricity in German electricity provision by 2050.

Political actors have increasingly demanded scientific expertise on how to pursue the "Energiewende", which has led to a variety of studies outlining long-term, model-based mitigation scenarios. These studies illustrate technology pathways for the German energy system that are consistent with the Government targets. Since the starting point of the scenarios constitutes the target formulations for the year 2050, they are normative scenarios in the sense that they specify which developments *ought* to occur for attaining the targets. The scenarios are based on numerical energy-system models with different scopes and scale, each relying on a variety of assumptions.

In order to synthesize the answers that existing mitigation scenario studies provide for the questions above, this paper pursues a comparative meta-analysis of selected mitigation scenarios that are consistent with the targets of the "Energiewende", see Table 1. WWF (2009) was commissioned by the non-governmental organization World Wildlife Fund (WWF) Germany and prepared by the Prognos AG. EWI/GWS/Prognos (2010) outlines a set of mitigation scenarios that analyze different prolongation periods for the nuclear phase-out and was commissioned by the Federal Ministry for Economics and Technology to inform the Government's long-term energy strategy known as the "Energy Concept" (Federal Government, 2010). The "Lead Study" scenarios in DLR/IWES/IFNE (2010; 2012) were commissioned by the Federal Ministry for the Environment and Nuclear Safety and served as a basis for the German National Renewable Action Plan (NREAP) required by the European directive 2009/28/EC. SRU (2010) was commissioned by the German Advisory Council on the Environment (SRU) and prepared by the German Aerospace Center (DLR) in collaboration with the SRU. Aside from SRU (2010), which focuses on the electricity sector alone, all the studies consider the German energy system as a whole. All four studies were relevant to the political debate (Fink and Burck, 2009). In addition, we compare these studies to our own model-based analysis (Schmid and Knopf, 2012), which was prepared over the course of the EU research project 'Engaging Civil Society in Low Carbon Scenarios'.

The specific questions this paper addresses are: Which combinations of strategic options for reaching a share of at least 80% renewables in the German electricity sector by 2050 do the scenario projections suggest? And what are the costs of the transformation? As model-based scenarios require numerous explicit and implicit assumptions, we identify and discuss several underlying assumptions that seem to be crucial, indicated not least by the controversial public and academic debates they have received. The investigation reveals several unanswered research questions that future research needs to tackle for providing sound scientific advice to the process of developing robust energy strategies for Germany.

Table 1. Overview of mitigation scenarios considered in this meta-analysis. * indicates relative to 1990.

Study	Scope	Target Formulation	Selected Scenarios
WWF (2009)	Germany	95% greenhouse gas emission reduction in 2050*	Innovation scenario [Inn] and variant [Inn_CCS] allowing for carbon capture and sequestration (CCS)
EWI/GWS/Prognos (2010)	Germany and Europe	40 / 85% CO ₂ emission reduction in 2020* / 2050*, ≥50% renewables in primary energy demand of 2050	[A1] , the scenario with the lowest extension of the nuclear-phase out of 4 years (until 2026)
DLR/IWES/IFNE (2010)	Germany and Europe+	80% CO ₂ emission reduction in 2050*, RES-E share ≥ 50% / 65% / 80% in 2030 / 2040 / 2050,	Scenarios [A] and [B-100%-S/H2] , which assume a share of 33% [A] and 66% [B] of electro-mobility in 2050's motorized individual transport mileage
DLR/IWES/IFNE (2012)	North Anta	RES-E share ≥ 100% in 2050 for [B-100%-S/H2]	Updated version of scenario [A], assumes 50% share of electro mobility
Schmid and Knopf (2012)	Germany	Maximize intertemporal welfare under a CO ₂ emission budget of 16 Gt over the period 2005-2050	Paradigm shift scenario [PS] and its variant [PS+] , which additionally allows for CCS and large-scale biofuel production
SRU (2011)	Germany and Europe+ North Africa	100% RES-E share in 2050 (analysis of the electricity sector only)	Scenarios [2.1.a] and [2.1.b] , which differ in assumptions on electricity demand in 2050 (500 vs. 700 TWh); both impose a zero net-import balance

This paper is structured as follows: Section 2 gives an account of the modeling methods applied in generating the selected scenarios. Section 3 analyzes the scenario projections on the strategic options for the long-term transformation of the German electricity sector to achieve high shares of RES-E and investigates the corresponding costs. Section 4 discusses underlying scenario assumptions that are critical for realizing the scenario projections and briefly indicates the policy leeway for the German Government. Section 5 summarizes and concludes.

2 Applied Modeling Methods

What are the applied methods in the studies for generating the scenario projections in view of the three strategic options for increasing the RES-E share in the German electricity sector? Whilst all the studies base their scenarios on numerical energy system models, they differ significantly with respect to how domestic RES-E generation, electricity demand and RES-E inputs are determined. The methods range from simply imposing exogenous assumptions on their development to having them determined as an endogenous model result. The latter method is the preferable in terms of scientific rigor. The ideal model for developing consistent mitigation scenarios for Germany would represent

all relevant systemic processes and their interplay and would generate coherent scenario projections that then could be justified by resorting to the causality structure of the model. However, due to numerical constraints resulting from computer power, solution algorithms as well as methodological challenges (e.g. unobservable data, lack of theoretical understanding of underlying processes) no such model is available to date. Each study pursues different means of coping with these challenges and combines the practices of exogenously assuming developments with endogenous modeling. A common variant is sequentially applying partial models that each focus on different aspects of the energy system and using output of one model as an input to another one. This practice, referred to as soft-coupling, is advantageous when focusing on technological details, but comes at the cost of impeding certain systemic feedback. The following describes the modeling methods applied in the studies for determining domestic RES-E generation, electricity demand and RES-E inputs (Table 2) – to the extent that they are reported in the publications – and highlights key implications of selected explicit or implicit assumptions.

Table 2. Overview of how electricity demand, RES-E capacities and RES-E imports are determined in the studies. Grey
shading indicates exogenous assumptions.

Study	RES-E capacities	Electricity Demand		RES-E Imports
WWF (2009)	Exogenous model input (to dispatch model), taken from DLR (2008)	Model result (from bottom-up simulation)	Model input (to dispatch model)	Residual value
EWI/GWS/Prognos (2010)	Exogenous model input (to dispatch model), determined by model of renewable energies	Model result (from bottom-up simulation)	Model input (to dispatch model)	Model result (from dispatch model)
DLR/IWES/IFNE (2010; 2012)	Exogenous model input (to "quantity framework"), based on extrapolation	Model result (from "quantity framework")		Model result for selected years (from EU electricity sector model)
Schmid and Knopf (2012)	Model result (from integrated welfare maximization)			Not considered in model, thus assumed to equal zero
SRU (2011)	Model result for 2050 (from EU electricity sector model), Interpolation before	Exogenous model input (to EU electricity sector model)		Assumed to equal zero in selected scenarios

In order to determine the domestic RES-E generation along with the annual changes in the installed RES-E capacities, the studies rely either on detailed dispatch modeling, optimization methods, or simply spreadsheet calculation. Both WWF (2009) and EWI/GWS/Prognos (2010) apply a dispatch

model to determine the cost-minimal dispatch of power plants for covering residual load, i.e. the demand time series minus priority feed-in of fluctuating RES-E technologies. Hence, by construction both RES-E capacity deployment and the associated feed-in must be determined exogenously before running a dispatch model. RES-E capacities are also exogenous to the "quantity framework" applied in DLR/IWES/IFNE (2010; 2012), which is in fact a spreadsheet simulation tool. None of these publications provide in-depth information on the rationale behind the selected RES-E capacity deployment and RES-E feed-in projections. For instance, WWF (2009) relies on the RES-E projections of DLR (2008), which is an earlier version of DLR/IWES/IFNE (2010; 2012). EWI/GWS/Prognos (2010) state: "The scenario construction of the electricity generation sector is further based on a model of renewable energies in Europe that represents several RES-E technologies in a regionally differentiated manner" (p. 28). It remains unclear how particular technologies and their deployment levels were chosen. DLR/IWES/IFNE (2010) state: "The capacity deployment path of RES-E technologies results from an extrapolation of the historical dynamics, under the assumption of priority feed-in for RES-E technologies until 2050." (p. 46). Here, the technical viability of the chosen technology mix is validated for selected years with the simulation tool SimEE. In Schmid and Knopf (2012) and SRU (2011), the installed RES-E capacities and feed-in are both determined endogenously through optimization methods. Schmid and Knopf (2012) apply the hybrid energy-economy model REMIND-D (Schmid et al., 2012). This model adopts a social-planner perspective with perfect foresight and determines the welfare-optimal domestic RES-E deployment pathway and average annual capacity factor for Germany in five-year time steps. For the year 2050, SRU (2011) applies the European and North African electricity sector model REMix (Scholz, 2010) to determine the costoptimal dispatch of RES-E technologies imposing a target share of RES-E of 100%. The deployment path between 2010 and 2050 was derived through interpolation. Therefore, nothing can be said about the optimality of the transitional RES-E capacity deployment in SRU (2011).

The approaches for determining future electricity demand range from detailed bottom-up representations over an economic top-down production function to treating demand exogenously by relying on projections of other publications. The first extreme is pursued in both WWF (2009) and EWI/GWS/Prognos (2010), which apply the same detailed bottom-up simulation model of the Prognos AG. It consists of a variety of sectorial sub-modules that generate differentiated projections of industrial, residential and commercial energy demands. One of the model outputs is the annual electricity demand along with its temporal load profile. The "quantity framework" applied in DLR/IWES/IFNE (2010; 2012) is a spreadsheet simulation tool that is similarly bottom-up oriented, but not as detailed as the Prognos demand model. The model REMIND-D used in Schmid and Knopf (2012) determines energy demand endogenously in a top-down approach by means of a calibrated

and parameterized production function. Here, electricity demand can respond to changes in the energy system, however, the results depend on the chosen elasticities of substitution. Finally, SRU (2011) consider electricity demand as entirely exogenous input for the REMix model and choose trajectories based on the upper and lower boundary of the range projected in German scenario literature.

Finally, the determination of RES-E imports primarily depends on the regional scope of the applied model(s). In WWF (2009) imports are a residual after considering demand, exogenous domestic RES-E generation and cost-minimal thermal electricity generation. In EWI/GWS/Prognos (2010) the dispatch model has 12 regions and calculates cost-optimal imports and exports. In DLR/IWES/IFNE (2010; 2012), it is not entirely clear how imports are determined. Nevertheless, they are validated with the European electricity sector model REMix that establishes cost-minimal generation and the associated import and export flows for each country. Schmid and Knopf (2012) abstract from imports since REMIND-D is a closed-economy model that consists only of the region Germany. In the selected scenarios of SRU (2011), electricity exchange with neighboring countries is allowed for over the course of the year; however, net electricity imports are imposed to equal zero.

This section has shown that even though all scenarios in this meta-analysis are based on numerical modeling, their projections are partly exogenous assumptions and partly endogenous model results. An important implication of the observation that the expansion of domestic RES-E capacities is assumed exogenously in the majority of the studies is that cost considerations are not necessarily their driving factor (cp. Edenhofer et al., under revision).

3 Comparative analysis of Scenario Projections

Despite that the scenario projections are based on very different modeling approaches, this section provides a comparative analysis thereof. On theoretical grounds this implies a strict "model democracy" in the sense of valuing model-based scenario projections equally regardless of the quality of input assumption and analytical merits (cp. Knutti, 2010), which Section 4 will focus on.

3.1 Strategic Options for Transforming the Electricity Sector

Which combinations of strategic options do the scenarios propose for reaching a share of at least 80% renewables in the German electricity sector by 2050? Before analyzing the scenario projections for each strategic option individually, Figure 2 illustrates the share of domestic RES-E generation in

total German electricity consumption/production (depending on the data reported) over time. The crosses indicate the Government Targets as formulated in §1 of the Renewable Electricity Act, which additionally allow for RES-E imports. Those scenarios that are below the cross in 2050 are still consistent with the target as they consider imports in this year, as shown below. An interesting finding is that the scenarios of Schmid and Knopf (2012) and SRU (2010) that do not consider any imports display a significantly faster acceleration in the share of RES-E over the next two decades and reach the Government's 80% minimal target share as early as 2025 to 2035. Four scenarios achieve a RES-E share of 100% in 2050. The scenarios of WWF (2009), EWI/GWS/Prognos (2010) and DLR/IWES/IFNE (2010; 2012) follow almost the same linear trajectory just above the Government's targets until 2040 that corresponds to a linear increase extrapolating the recently observed growth rate: 15 percentage points per decade.

Figure 3 displays the scenario projections on domestic RES-E generation together with historical data. The scenario projections fan out significantly over the long-term: Whilst the spread in RES-E generation amounts to 200-350 TWh in 2020 already, it increases up to as much as 250-700 TWh in 2050. Again, the scenarios that do not consider imports display an accelerated increase in the growth of domestic RES-E generation compared to those ones that do consider imports. The implications of the different expansion strategies would be severe: Whether RES-E generation ought to increase by factor two or five over the coming four decades would make a considerable difference for the design and volume of required RES-E support schemes and, correspondingly, on the future size of the domestic RES-E market. Pursuing a high RES-E expansion strategy would require a doubling of the domestic RES-E market already in the coming decade and thus called for timely action. A low RES-E expansion strategy allowed three additional decades time for the same development.



Figure 2. Share of domestic RES-E generation in total German electricity consumption/production (depending on the data provided). Includes historical data (BMU, 2012; BMWi, 2012) and projections from a selection of model-based, long-term mitigation scenarios, see Table 1. Crosses indicate the official minimum targets shares of RES-E in electricity provision of §1(2) of the Renewable Energy Sources Act (that additionally allow for RES-E imports)



Figure 3. Electricity generation from domestic RES-E. Includes historical data (BMU, 2012) and projections from a selection of model-based, long-term mitigation scenarios, see Table 1.

Which technologies are employed for domestic RES-E generation in the different scenarios? Figure 4 presents the cumulative RES-E generation by technology over the period 2010-2050, excluding

hydropower because the limited domestic potential is already exploited. Figure 4 reveals that electricity generation from wind is the most important pillar of the future technology mix in all the scenarios. Together, onshore and offshore wind contribute 2740-11550 TWh (55-70%) of cumulative domestic RES-E generation between 2010 and 2050. With the exception of the SRU (2011) scenarios, in which offshore wind is dominant, the two technologies contribute in roughly equal parts. Electricity generation from biomass provides 1490-2360 TWh (10-30%) across the scenarios. Biomass is a dispatchable RES-E technology that is important for balancing fluctuations from the variable technologies wind and solar photovoltaic (PV). However, the potential of biomass is limited mainly due to ecological concerns (Nitsch et al., 2004). Solar PV, the relatively most expensive technology in terms of levelized cost of electricity today (Fischedick et al., 2011), ostensibly does not play a major role in either of the scenarios with 720-2870 TWh (3-16%) of cumulative domestic RES-E generation. Finally, geothermal electricity generation only plays a role in those scenarios that refrain from RES-E imports contributing 0-2430 TWh (0-13%). The assumed reason is that it is a high cost technology that is only employed when cheap imports are not available. Hence, in relative terms the scenarios suggest that wind offshore and onshore are the most important technologies for domestic RES-E generation, closely followed by biomass, and to a lesser extent solar PV as well as geothermal.



Figure 4. Cumulative domestic RES-E generation over the period 2010 to 2050. Includes projections from a selection of model-based, long-term mitigation scenarios, see Table 1.

Since the share of domestic RES-E generation increases steadily in all the scenarios, those which project a relatively low expansion of domestic RES-E generation are likely to project a relatively lower trajectory for electricity demand. Figure 5, which plots normalized electricity demand with the base year 2010, confirms this intuition. The WWF (2009) and EWI/GWS/Prognos (2010) scenarios project reductions in electricity demand of 25%-35% in 2050 compared to 2010. According to the scenarios, an energy strategy that relies on a comparatively low expansion of domestic RES-E generation must simultaneously induce a decisive turnaround in energy efficiency trends: German electricity demand has increased by 20% over the past two decades, although it has stagnated in recent years. However, this development is attributed to the global financial crisis rather than to dedicated efficiency policies (BMWi, 2010). Such a deliberate energy-saving strategy is also favored by the German Government, which aims to achieve a 10% reduction in electricity demand in 2025 and 25% in 2050 relative to 2008 in its long-term "Energy Concept" (Federal Government, 2010). It is necessary to acknowledge that substantial improvements in energy efficiency improvements in the residential and industrial sectors are not only assumed for scenarios that display a decreasing electricity demand in Figure 3, but also to a certain extent for those that remain stable over time. This is due to two developments: On the one hand, in the transport sector electricity demand is assumed to increase, and on the other hand GDP is projected to grow continuously in all scenarios. Thus, a stable or slightly decreasing electricity demand is tantamount to a decoupling process of electricity consumption from economic growth or, in other words, increasing energy efficiency. The strong upward trend in electricity demand in the scenario DLR/IWES/IFNE (2010) [B100%] between 2040 and 2050 is due to the assumption that by 2050, 41% of the domestically generated and imported electricity is converted to hydrogen in power to gas facilities. The hydrogen is either used in the transport sector as fuel or serves as chemical long-term storage (DLR/IWES/IFNE, 2010, p. 80).

The strategic option of RES-E imports is exploited by all the scenarios that have the option to do so (Figure 6). This would imply that Germany turned from a net exporter of electricity towards a net importing country with 50-200 TWh in 2050. The scenarios assume that this imported electricity is produced from RES-E in European countries; DLR/IWES/IFNE (2010; 2012) additionally allow for imports from North Africa. EWI/GWS/Prognos (2010) in principle also consider nuclear electricity imports but the amounts are not explicitly mentioned. The share of German electricity demand that would be satisfied by imports in 2050 ranges from 11% in the scenario DLR/IWES/IFNE (2012) [A] to considerable 25% in the scenario DLR/IWES/IFNE (2010) [B-100%].



Figure 5. Normalized electricity demand. Includes historical data (BMWi, 2012) with [2010=1] and projections from a selection of model-based, long-term mitigation scenarios with [base year = 1], see Table 1.



Figure 6. Electricity import balance. Includes historical data (BMWi, 2012) and projections from a selection of model-based, long-term mitigation scenarios, see Table 1. Imports are assumed to be produced with RES-E technologies.

To synthesize the findings of this subsection, the investigation has revealed that the scenarios exploit the strategic options of (i) increasing domestic RES-E generation, (ii) decreasing electricity demand and (iii) increasing RES-E imports to substantially different extents, albeit with some

inherent pattern. In order to visualize this pattern, Table 3 provides a stylized comparative account of the extent to which the scenarios emphasize each strategic option by rating them based on three categories. The relative rating is obtained by dividing the spread of the scenario projections for each option in 2050 by three and then attributing +, ++ or +++ to the scenarios, indicating for a scenario that it falls into the bottom, middle or upper third. Note that for electricity demand, the outlier scenario DLR/IWES/IFNE (2012) [100%-SH2] scenario was omitted for calculating the spread in 2050, as it is the only that assumes large-scale hydrogen production from RES-E.

The scenarios cluster in three groups based on the combinations of strategic options employed. The first group heavily relies on exploiting energy efficiency potentials to reduce electricity demand, which is satisfied by relatively low domestic RES-E generation and moderate RES-E imports. The second group refrains from imports and balances moderate to high domestic RES-E generation against high to moderate reductions in electricity demand. A similar focus on domestic RES-E generation is found in the third group, in which relatively higher electricity demand from the electrification of the transport sector and hydrogen generation from RES-E is balanced with RES-E imports. Overall, the scenarios suggest that the long-term target of reaching a share of at least 80% RES-E in the German electricity sector can be achieved with distinct combinations of the three strategic options.

Scenario	Domestic RES-E	Demand Reduction	RES-E Imports
WWF (2009) [Inn_CCS]	+	+++	+
EWI/GWS/Prognos (2010) [A1]	+	+++	++
WWF (2009) [Inn_noCCS]	+	+++	+
DLR/IWES/IFNE (2012) [A]	++	++	+
SRU (2011) [2.1.a]	++	+++	
Schmid and Knopf (2012) [PS]	++	++	
Schmid and Knopf (2012) [PS+]	++	++	
SRU (2011) [2.1.b]	+++	+	
DLR/IWES/IFNE (2010) [A]	++	+	++
DLR/IWES/IFNE (2010) [B-100%]	+++	+	+++

Table 3. Stylized comparative account of the extent to which the scenarios exploit each of the three strategic options for transforming the German electricity sector towards high shares of RES-E.

3.2 Costs of the Transformation

What are costs for transforming the German electricity system towards high shares of RES-E as suggested by the mitigation scenarios? Conceptually, it is important to consider the following cost factors for assessing total transformation costs: (i) Investment costs and operation and maintenance costs for domestic RES-E capacities, (ii) system integration costs to accommodate the temporal and spatial variability of fluctuating RES-E feed-in and (iii) costs for implementing energy efficiency measures (Pahle et al., 2012). It is not possible to perform a structured comparative analysis of the projected costs of the transformation for at least three reasons. First, the studies use different approaches to quantify costs due to the different modeling approaches (cp. Pahle et al., 2012). Second, the studies do not consider all individual cost factors, and, third, costs are reported only selectively. Regarding the approaches to quantify costs, it is common practice to either pursue a top-down analysis of macroeconomic costs induced by the transformation or a bottom-up analysis of costs accrued in the energy system (cp. Capros et al., 2010).

Four studies report the macroeconomic costs of the transformation scenarios compared to a reference scenario; however, both the metrics and the characteristics of the reference scenarios differ significantly. WWF (2009) finds that the [Inn] scenario incurs net additional costs of 0.3% of GDP over the time horizon 2010-2050, relative to a reference scenario that by 2050 achieves a RES-E share of 56% and CO₂ emission reductions of 50% relative to 1990. The scenarios [PS] and [PS+] of Schmid and Knopf (2010) indicate higher cumulative discounted GDP losses of 1.4% and 0.8% over the period 2005-2050, respectively. This is despite the fact that the respective reference scenarios only reduces CO_2 emission by 40% relative to 1990, albeit reaching RES-E shares of 65% and 63%. In contrast, EWI/GWS/Prognos (2010) find that the scenario [A] leads to a GDP that is 0.6% higher in 2050 compared to the reference scenario characterized by 62% CO₂ emission reduction and a 54% RES-E share in 2050. The major reason is the drastic increase in energy efficiency that leads to a comparatively lower primary energy demand and correspondingly to savings in primary energy imports. DLR/IWES/IFNE (2010; 2012) also report cost advantages of the mitigation scenarios [A] accruing to €665 billion and €570 billion in 2050 – in terms of "system-analytic difference costs", i.e. when compared to a fictitious electricity supply based on fossil and nuclear energy that refrains from the use of renewables. The description of the reported metrics makes it clear that it is not possible to compare them in a sensible way. Since all four studies analyze the transformation of the German energy system as a whole, these cost figures cannot explicitly be related to the transformation of the electricity sectors. After all, a comprehensive answer to the question of the macroeconomic costs of the transformation of the German electricity sector is currently missing.



Figure 7. Cumulative RES-E capacity investments and total RES-E electricity generation over the period 2011-2020. The numbers are not discounted, as they were made available to the authors in this manner.

The bottom-up perspective on transformation costs offers itself somewhat more accessible to analysis. Figure 7 illustrates undiscounted cumulative RES-E investments between 2011 and 2020 (the data was made available to the authors in aggregated manner only). The scenarios suggest that the total cumulative investment costs for RES-E capacities over the next decade range between €75-210 billion. In all scenarios approximately half of the investments are directed into solar PV, the other half in wind capacities and a minor share to biomass and geothermal. Considering that across scenarios solar PV delivers only 3-16% of cumulative electricity generation between 2011 and 2050 (cp. Figure 4), the projected investments into solar PV over the next decade appear disproportionate. This is also the case for the de facto investments of 2010 and 2011 in which solar PV had a share of 77% (BMU, 2011; BMU, 2012). The disproportionate share of solar PV in historical investments is due to the fact that the specific investment costs of solar panels have continuously decreased by more than 60% since 2006 (BSW Solar, 2012), but feed-in tariffs have been reduced only sporadically. This resulted in double-digit profit margins and solar-PV attracted as much as 35 Bn € of capital investments in 2010 and 2011 alone. The increasing cost burden induced by the guaranteed feed-in tariff for solar PV led to the adoption of an upper limit of 52 GW of solar PV capacity in the 2012 amendment of the Renewable Energy Sources Act.

Most of the scenarios did not project such a fast growth in solar PV capacity deployment, as becomes evident by the discrepancy between scenario projections and historical deployment data in

Figure 8. It reveals that nearly all scenarios underestimate the strong increase of solar PV deployment in the recent years, while for wind offshore it is the opposite, i.e. all models show increasing deployment while in reality only a few capacities are in place. The upper limit of 52 GW for solar PV support is indicated by the dashed line and coincides with the value in the DLR/IWES/IFNE (2010) scenario, which formed the basis for the German Government's National Renewable Action Plan (NREAP) submitted to the European Union in 2010 (DLR/IWES/IFNE, 2012). On the other hand, regarding wind offshore capacities the feed-in tariff so far led to only 200 MW connected to the grid in June 2012 (Deutsche Wind Guard, 2012). A further 600 MW of offshore projects are under construction, 4852 MW are approved and 1267 MW are under approval (bdew, 2012), adding up to 7.5 GW in projects. This is far below the tentative Government target of 25 GW by 2030 (Federal Government, 2010), which again coincides with the DLR/IWES/IFNE (2010;2012) scenarios. Thus, the scenarios suggest that investments into RES-E capacities should diversify from solar PV to other RES-E technologies, particularly wind offshore. Possible reasons for the discrepancies between model projections and de facto developments will be discussed in Section 4.1.



Figure 8. Installed capacities for solar PV (left) and wind offshore (right) until 2030. Includes historical data (BMU, 2013) and projections from a selection of model-based, long-term mitigation scenarios, see Table 1. The dashed line at 52 GW for solar PV is the upper limit fixed in the law of the renewable supporting scheme in Germany, see text.

The reported cost figures for operation and maintenance costs for RES-E capacities, system integration costs and costs for implementing energy efficiency measures are so limited that a comparison is not worthwhile. In fact, the majority of the scenarios take neither system integration costs nor energy-efficiency related expenses into account explicitly. This leads to the observation

that the classical triangle of energy policy goals – environmental protection, security of supply and cost-efficiency – is skewed by construction in the mitigation scenarios under analysis: While the environmental protection targets are prescribed in the scenario formulations and energy security is ensured by means of balance equations in the model, system cost minimization plays a subordinate role, particularly in those scenarios that treat RES-E capacity deployment as exogenous assumption.

4 Discussion of Underlying Scenario Assumptions

As regards a German energy strategy for the electricity sector that is consistent with the Government's targets, the previous Section identified the key policy levers: increasing domestic RES-E generation, energy efficiency and RES-E imports. This Section turns towards critically reviewing the scenarios' underlying assumptions concerning these three strategic options. We limit the discussion to those scenario assumptions that seem to be crucial, indicated not least by the controversial public and academic debates they have received. Further, we briefly indicate the leeway for policy intervention of the German Government.

4.1 Increasing Domestic RES-E Generation

The public debate on RES-E capacity expansion in Germany mainly revolves around the question of increasing cost burdens and cost-efficiency as well as potential security of supply issues related to the fluctuating feed-in of electricity generation from wind and solar (Schiermeier, 2013). Regarding these classical energy policy goals, the decisive assumptions in the scenarios are postulated investment cost reductions of RES-E technologies as well as a more or less implicitly assumed built-up of system integration measures to accommodate uncertain, fluctuating and spatially dispersed feed-in. The latter both serves to minimize system costs in the face of high share of RES-E feed-in as well as to ensure security of supply. The following subsequently presents and discusses the model assumptions regarding future reductions of investment costs and the deployment of system integration measures.

A core assumption in all of the studies with regard to system costs is that technological learning processes will occur when capacity deployment is incentivized either by support schemes or the market, leading to substantial reductions in investment costs. This notion of technological learning is referred to as learning-by-doing (cp. Junginger et al., 2010). It has been identified empirically as a statistical relationship between investment costs and cumulative installed capacity, quantified by a

parameter known as the learning rate. It expresses the rate at which specific investment costs decrease when cumulative capacity is doubled. For modeling purposes, regression estimates of learning rates are interpolated and either serve as an orientation to estimate cost reductions exogenously or as a direct input for the model, if learning is represented as an endogenous process, thereby influencing the future technology mix. Table 4 presents the learning rates that are reported in the scenario studies. As costs are an important factor, it is problematic that the learning rates are either reported only selectively or not at all in the scenarios that determine RES-E capacity expansion exogenously. Moreover, the choice of learning rates is usually not motivated explicitly.

Publication	Solar PV	Wind Onshore	Wind Offshore	Geo- thermal	Biomass
WWF (2009)	n.a.	n.a.	n.a.	n.a.	n.a.
EWI/GWS/Prognos (2010)	n.a.	n.a.	n.a.	n.a.	n.a.
DLR/IWES/IFNE (2010)	20%	n.a.	10%	n.a.	n.a.
DLR/IWES/IFNE (2012)	20%	n.a.	10%	n.a.	n.a.
Schmid and Knopf (2012)	20%	6%	12%	0%	0%
SRU (2011)	25.9%	11.5%	18.6%	n.a.	2.2%

Table 4. Learning rate assumptions of the different RES-E technologies as reported in the publications.

The application of the learning rate concept in the scenarios postulates a direct and fixed relationship between capacity deployment and cost reductions. Literature provides ample evidence that using learning rates for projecting RES-E investment costs is a method that suffers from serious shortcomings. The estimation of learning rates is highly sensitive to the timing of the underlying data both in terms of when the forecast was made and the duration of the data set (Nemet, 2009). Furthermore, while the explanatory power is generally high in learning rate estimations for the modular technologies such as solar PV (Junginger et al., 2008), it is very low in estimates for offshore wind, where little data exists (Neij, 2008; van der Zwaan et al., 2011). For offshore wind, specific investment costs have actually increased in recent years and future cost-reductions are assessed as uncertain (e.g. Heptonstall et al., 2012; Panzer, 2012). Hence, a learning rate of 10% - 19% for offshore wind as assumed in the scenarios may be challenging to realize. Besides assumptions on specific investment costs, a parameter that indirectly affects the costs of RES-E electricity generation in the scenarios is the assumption about full load hours that RES-E capacities can achieve. Average annual full load hours in energy system models are generally based on meteorological wind speed and solar irradiation data in combination with an average wind turbine and solar panel configuration to derive feed-in. However, realized full load hours are often significantly lower, particularly for wind

power. Boccard (2009) (2012) estimates that due to suboptimal siting induced by human and political economy factors, empirical full load hours are 20% lower in Germany than could be expected from meteorological wind speed data. In the future, improved configuration of wind turbines could alleviate this difference (Agora Energiewende, 2013a).

Whether the postulated cost reductions for RES-E deployment actualize in Germany depends on a variety of factors. These include future learning-by-doing effects both on a global and domestic scale, and how market values of RES-E feed-in and perceived technology-specific investment risks develop. Under current legislation any solar PV capacity additions above 52 GW will need to be profitable at market prices. However, the competitiveness of variable renewables decreases with RES-E penetration. They supply electricity at zero marginal costs, leading to low electricity prices during periods with ample wind and/or solar irradiation. This "self-cannibalizing" effect may render the competitiveness of large-scale wind and solar PV capacity deployment more difficult than is acknowledged to date (Edenhofer et al., under revision). Investment risks are not accounted for in the energy system models as they are deterministic models that do not take into account uncertainties besides those covered by different scenario definitions. It is implicitly assumed that sufficient financing can be acquired for RES-E capacity deployment.

The case of wind offshore shows that many real-world constraints are not taken into account in the models, particularly related to investment risks. While the technical potential of wind offshore is abundant in Germany, (IWES, 2012), wind offshore constitutes a centralized, high-risk and highmaintenance technology that is only worthwhile installing on a large scale, as opposed to the modular and low-risk technology solar PV. Solar PV is accessible to small investors, not least indicated by the fact that private persons and farmers pursued 51% of solar PV investments in 2011 (trend:research, 2011), an over proportional share as compared to the overall financing structure for RES-E capacities in Germany. A decisive growth in offshore wind deployment on the contrary requires tapping the capital resources of industry and institutional investors, or pooling private investments. However, returns on investment of wind offshore projects in the German waters are currently small and uncertain (Richter, 2009). In the North Sea, only far-shore projects are allowed due to the Wadden Sea National Park. Since increasing water depths and distance to the shore are the main factors influencing investment costs for offshore wind (Prässler and Schaechtele, 2012), projects in less challenging sites outside of the German seas are more interesting for profit-seeking investors. In fact, German far-shore sites are expected to deliver an acceptable return on investment only upon deployment of high yield wind turbines with 5 MW capacity (Zeelenberg and van der Kloet, 2007). To date, only 37 of these turbines have been installed worldwide, of which 29 are in

Germany (IWES, 2012). The fact that 5 MW turbines are still an immature technology discourages project developers, banks, insurances and financial investors (Richter, 2009). In the RES-E markets, proven reliability of a technology is found to be a necessary condition for investment (Masini and Menichetti, 2012). Many of these aspects are not considered in the models but are implicit assumptions for the realization of the strategic option of increasing RES-E deployment.

But high RES shares not only create a burden on costs – they also have an impact on security of supply. Increasingly high shares of wind and solar PV with their uncertain, fluctuating and spatially dispersed feed-in structure create challenges for system stability, which requires electricity supply to match demand at any time and place. Technical solutions for integrating wind and solar PV into the electricity system include an extension of (i) power grid infrastructure for transporting electricity to demand centers and large-area pooling of fluctuations, (ii) dispatchable generation capacities to provide short-term balancing as well as back-up capacities in low RES-E feed-in periods, (iii) demand side management (DSM) to adapt load to feed-in, reduce peak load, or provide ancillary services, and (iv) storage capacities to cope with fluctuations on both short and long time-scales, thereby avoiding the need for curtailment in high RES-E periods (Sims et al., 2011). Given that all the scenarios report high shares of wind and solar PV (cp. Section 3.2) the question arises of what they assume regarding system integration.

All studies assume at least some system integration options are deployed in the future; however, in most cases the costs are not accounted for (cp. Edenhofer et al., under revision). WWF (2009) assumes that storage capacities and demand-side management will be installed, which translates into higher average full-load hours of conventional, dispatchable power plants in the dispatch model (p. 18). The study remains vague on which storage technologies and DSM options are to be deployed and at what costs. Rather, they provide a gualitative discussion on these issues and frame it as an unresolved research question (p. 476). In EWI/GWS/Prognos (2010, p.187) it is assumed that DSM reduces peak load according to the potentials determined in dena (2010), yet the costs appear not to be included in the model. Whilst storage possibilities are limited to limited domestic pumped hydro storage potentials, the European dispatch model considers power grid expansions between individual countries. DLR/IWES/IFNE (2010; 2012) put a deliberate focus on validating that the exogenously determined RES-E capacity deployment is technically feasible by applying the simulation model SimEE. This model has an hourly resolution and assumes that all system integration options listed above are in place, including adiabatic compressed-air storage and the electrolysis of RES-E power to hydrogen (DLR/IWES/IFNE, 2010, p.95). However, the integration costs are not reported systematically and such an ex-post validation with a simulation tool can by

definition not reveal whether the particular configuration of the electricity system is cost-optimal. In Schmid and Knopf (2012) only the integration option of flexible back-up capacities represented explicitly in the model REMIND-D. System effects of uncertain and variable RES-E feed-in are modeled endogenously by a parameterized residual load duration curve approach (Ueckerdt et al., 2011). Finally, the REMix model applied in SRU (2010) endogenously determines European power grid expansion between countries, flexible back-up capacities and the deployment of the storage technologies pumped hydro, adiabatic compressed-air storage and power to hydrogen. While the degree to which system integration measures are considered in the scenarios differs substantially the common denominator is that they are assumed to be in place, be it implicitly or explicitly.

However, none of the options are currently deployed at a level sufficient to integrate rising shares of RES-E as projected by the mitigation scenarios into the German power grid (dena, 2010). In fact the stability of the German power grid was already critical in the winter of 2011/2012 (Federal Network Agency, 2012). In order to guarantee stability and security of electricity supply in the face of substantially rising RES-E shares the deployment of an optimal portfolio of system integration measures needs to be incentivized, which constitutes a considerable challenge and requires decisive institutional reforms as well as fundamental technological progress. Publicly discussed measures include e.g. the speeding up approval procedures for power grid extensions (dena, 2010), reforming the market design of control power markets (Federal Network Agency, 2012) and re-designing the pricing system for end customers in a more flexible fashion to enable DSM (Sims et al., 2011). The only short- and long-term storage technology that constitutes a profitable investment today is pumped hydro storage but Germany's pumped hydro potential is too small to do the job alone (ETG Task Force Energiespeicher, 2008). All other available technologies are immature and still require considerable effort in research and development to enable their large-scale deployment (ETG Task Force Energiespeicher, 2008). Thus, the deployment of system integration measures needs to accelerate in the mid-term future in order to both ensure security of supply and keep system costs low.

Overall, the policy leeway for the German Government for increasing domestic RES-E generation is rather large, even though numerous challenges need to be tackled on the route. Ultimately the question is at which costs the expansion can be realized and, how these costs are distributed among electricity consumers and whether sufficient investments into RES-E capacity expansion can be stimulated. Among the key recommendations of the International Energy Agency (IEA) are that additional policy measures are needed in order to ensure cost efficiency, effectiveness and a fair distribution of the costs of the Energiewende (IEA, 2013). A possible way of reducing costs could be

to put less emphasis on offshore wind and more on onshore wind, which is comparably more costefficient (Agora Energiewende, 2013b). Other strategies to keep transformation costs as low as possible could be to avoid the need of excessive domestic RES-E capacity expansion by reducing overall electricity demand through exploiting energy efficiency potentials and importing electricity generated by RES-E capacities in more cost-efficient sites outside of Germany. These options will be reviewed in the following.

4.2 Energy Efficiency

The strategic option of decreasing electricity demand through leveraging energy efficiency potentials is particularly important in the scenarios of WWF (2009) and EWI/GWS/Prognos (2010), which apply a bottom-up model for determining electricity demand. Bottom-up models adopt an engineering perspective by incorporating detailed descriptions of technologies while assuming market adoption of the most efficient technologies (Hourcade and Robinson, 1996). These models postulate that market forces do not operate perfectly and frequently identify an efficiency gap between current technology penetration and the best available techniques. Suggested policy implications of bottom-up models thus are mainly to remove barriers to adoption of the best available technique. In the global energy system models literature, bottom-up models are found to be highly optimistic in terms of technical mitigation potential, due to picking "low-hanging fruits" that may even come at negative price that are in fact not picked today (Grubb et al., 1993; Hourcade and Robinson, 1996). In those scenarios that do not apply a bottom-up model, the rationale behind energy efficiency improvements is not specified further. Instead, electricity demand is simply assumed to follow the postulated trajectory.

Although energy efficiency improvements are a crucial element of all scenarios to a different extent, it constitutes a controversial policy domain in which it is highly unclear whether ambitious targets can be realized. A long-standing debate in literature has observed numerous policy puzzles, the most prominent being the rebound effect. It postulates that an increase in the energy efficiency of a specific energy service may lead to a direct increase in the demand for that service or indirectly increase the demand for other energy-intensive services and, ultimately, increase net energy consumption (Sorrell et al., 2009). Attempts to measure rebound effects are subject to numerous methodological and data measurement problems, however, a recent literature survey suggests that the direct rebound effect may reduce the energy efficiency projections of bottom-up models by as much as 30% (Sorrell, 2007). In any case, rebound effects should be taken into account when designing energy efficiency policies. They may be mitigated through carbon taxes or mitigation caps,

i.e. through higher prices on carbon-intensive energy (Sorrell, 2007). A further policy puzzle is that many proposed energy efficiency policies are in fact energy-saving policies that aim to force the incumbent, regulated utilities to implement energy efficiency programs that will after all reduce the demand for their product (Brennan, 2011). Alternative business models seem necessary to exploit energy efficiency opportunities, such as energy contracting. In Germany, one company began offering energy contracting for consumers, but could not establish themselves on the market and withdrew their activities in the end-customer segment in order to focus solely on organizational and institutional customers (Kofler Energies Power AG, 2011).

Behavioral changes are implicitly assumed in the models as it is not possible to model these processes explicitly in technology-focused models that do not consider individual agents. However, literature provides evidence of a collection of market and behavioral failures that intend to explain the difficulties in promoting energy efficiency observed in the past (e.g. Sorrell et al., 2004; Gillingham et al., 2009; Thollander et al., 2010). Many of the identified market failures (e.g. environmental externalities, average-cost electricity pricing, liquidity constraints, R&D and learning-by-doing spillovers) are not unique to energy efficiency, thus they call for a broader policy response including carbon pricing, innovation policies and electricity market reforms. Information and behavioral failures, such as lack and asymmetry of information, principal-agent problems, split incentives, hidden costs or bounded rationality on the other hand call for more specific energy efficiency policies (Gillingham et al., 2009).

Current policies that are targeted at improving the efficient use of energy in Germany have been mainly spurred by European legislation. The "Second National Energy Efficiency Action Plan" (NEEAP) (BMWi, 2011) outlines the mix of policy instruments by means of which the German Government intends to meet the requirements of the European energy efficiency directive (European Parliament and the European Council, 2009). As regards electricity, the major instruments are financial support programs, minimum efficiency requirements and energy labels for appliances, an eco-tax on electricity, self-voluntary programs of industry and information campaigns. However, despite these instruments being in place for the last decade, absolute German electricity demand did not decrease significantly (cp. Figure 5). Having assessed energy efficiency policies and measures in Germany, Schlomann and Eichhammer (2012) conclude that in order to meet the ambitious energy efficiency targets of the German government additional efforts as compared to the second NEEAP are necessary. As the toolbox of policy instruments is not yet exploited there is still substantial leeway for incentivizing the relevant actors and institutions towards a more efficient use of electricity.

4.3 **RES-E Imports**

Importing RES-E electricity from other European or even North African countries where RES-E generation is more cost-efficient due to more favorable wind speed or solar irradiation sites is a silver bullet in cost-optimizing energy system models. A prominent example are the DLR/IWES/IFNE (2010; 2012) scenarios for which REMix model finds it optimal to significantly increase European grid capacities and import up to 25% of German electricity demand in 2050. In the SRU (2010) scenarios net imports are zero in 2050, however, the large hydro reservoir capacities of Norway are employed to balance fluctuations arising from fluctuating RES-E feed-in in Germany and central Europe. Such model results crucially hinge on the obvious assumptions that (i) RES-E production in non-German countries is actually more cost-efficient, (ii) sufficient power grid capacities to transport the electricity is available and (iii) RES-E capacity deployment starts to increase in these countries as postulated by the models. Moreover, implicit assumptions are that European and/or North African actors cooperate to institutionalize a truly integrated electricity system with shared responsibility for energy security issues and similarly high target shares for RES-E generation.

While the vision of a truly integrated pan-European electricity system promises resource- and costefficiency, there are numerous obstacles that yet need to be tackled for paving the road. The decisive assumption that RES-E generation in sites with optimal natural resources in non-German countries is actually more cost-efficient is challenged, e.g. by Dinica (2011), who argues that production costs of RES-E technologies are not only influenced by technology-specific factors, but also technology-complementary, institutional and resource-related factors that are likely to increase with increasing domestic capacity deployment. Such cost factors, e.g. licensing fees, are usually not explicitly taken into account in energy system models, and if so, they are expected to decrease rather than increase in the course of future technology learning. While the expansion of the European power grid is coordinated by the European Network of Transmission System Operators for Electricity (ENTSO-E), the capacity expansion might not be as fast as postulated by the mitigation scenarios. In the past years, transmission projects of pan-European significance have been challenged, significantly delayed or even jeopardized because of a lack of social acceptance, lengthy permitting procedures, difficulties in acquiring financing and inherent uncertainties of predicting the future location of generation capacities, particularly RES-E capacities (ENTSO-E, 2010).

Klessmann et al. (2011) evaluate the current status of renewables deployment, policies and barriers in the European Member States and conclude that the de facto built-up of RES-E capacities in the different Member States is on aggregate beyond the self-imposed quantities as specified in the National Renewable Action Plans (EEA, 2012) and call for increased policy effectiveness in countries that lag behind to close the gap to the top runner countries. The recent "Renewable energy progress report" (European Commission, 2013b) comes to the same conclusion that further efforts will still be needed from the Member States in order to reach the European 2020 targets (20% greenhouse gas emission reduction relative to 1990, 20% share of renewables in final energy demand and 20% energy efficiency improvement). This emphasizes that a truly integrated European electricity system requires fostering and triggering technical, institutional and political developments in a coordinated manner. German policymakers can clearly not determine this alone but can positively influence the negotiations in the direction of a more integrated European electricity system. The current negotiations on a 2030 framework (European Commission, 2013a) are an opportunity that should not be missed, see e.g. Geden and Knopf (2013).

5 Summary and Conclusion

This paper performed a meta-analysis of ten model-based mitigation scenarios for Germany, taken from six recent studies to address the following questions: Which combinations of strategic options for reaching a share of at least 80% renewables in the German electricity sector by 2050 do the scenario projections suggest? And what are the costs of the transformation? The main findings regarding the first question are: (i) In order to attain the long-term target of at least 80% renewables in the German electricity sector, the scenarios exploit the strategic options of domestic RES-E generation, electricity demand reductions and RES-E imports to substantially different extents, (ii) domestic RES-E generation increases in all scenarios with onshore and offshore wind representing the most important technologies, (iii) the scenarios that decisively exploit the strategic option of reducing electricity demand require a relatively low expansion of domestic RES-E generation, and (iv) some models rely heavily on electricity imports from other European Member States. The magnitude and sign of the costs of the transformation remain vague and a structured comparative analysis across scenarios is impeded by the varying methodical approaches to quantify costs, failure to consider all individual cost factors and selective reporting of costs. Furthermore, the reference scenarios used to determine the incremental costs range from a moderate transformation scenario to one with a strong deployment of nuclear and fossil fuels without any renewables. Generally, the analysis of costs appears to play a subordinate role in the scenarios.

The discussion about crucial underlying assumptions of the scenario projections revealed the following recurring themes: It is unclear whether (i) RES-E and system integration technology development will be as cost-competitive as postulated, (ii) the implicitly assumed institutional requirements will be realized, and (iii) the relevant actors in the transformation process will be

incentivized accordingly. With respect to the strategic option of increasing domestic RES-E generation, it is unclear whether sufficient financing can be acquired, which is closely linked to how investment risks develop, and whether the necessary system-integration measures will be in place. For the strategic option of decreasing electricity demand, the most daunting question is how to overcome the numerous market failures and policy puzzles in order to incentivize the users of electricity-based energy services to leverage energy-efficiency potentials. Finally, the viability of the strategic option to import RES-E import is uncertain and largely depends on whether policy targets in European Member States will be aligned in order to cooperatively aim for higher shares of RES-E.

It deems that for implementing the strategic options suggested by the mitigation scenarios, technology development, institutional settings and the governance of relevant actors need to be aligned. This finding is perfectly in line with the theoretical insights of literature that intends to characterize the nature of the energy system, arguing that energy systems are best perceived as socio-technical systems in which the elements of technology, institutions and actors play distinct but deeply intertwined roles (e.g. Kroes, 2006; Unruh, 2006; Reichel 2012). A strategy that intends to transform a socio-technical system successfully thus needs to address all elements of the system in a coherent way; else the strategy is likely to encounter barriers to implementation and possibly induces system-wide effects that might even contradict its intended aim. At large, the development of such robust energy strategies is an urgent task while at the same time a formidable challenge for the German policymaker. The process of developing such strategies could benefit from sound scientific advice in several ways that can be deduced from the discussion above and that could feed into a "social learning process" (cp. Edenhofer et al., under revision).

A better understanding of energy technology-diffusion processes from a socio-technical systems perspective could provide a more accurate picture of the actors involved and the institutions required. This applies to technologies on all levels of the electricity system, from supply-side technologies used to generate, distribute and store electricity to demand-side technologies that convert the final energy electricity into energy services. Once a clearer picture of the dependencies in the system emerges, it will be possible to give a more comprehensive account of the prerequisites and consequences of the multiple strategic options for transforming the energy system as well as the nature and magnitude of the corresponding costs – and opportunity costs. Such an assessment could provide a sound scientific basis for informing the German quest to succeed in its self-imposed experiment known as the "Energiewende".

Acknowledgements

The authors would like to thank Ottmar Edenhofer, Falko Ueckerdt and Philippe Quirion for helpful comments.

References

Agora Energiewende (2013a). Entwicklung der Windenergie in Deutschland. Eine Beschreibung von	
aktuellen und zukünftigen Trends und Charakteristika der Einspeisung von	
Windenergieanlagen KURZSTUDIE. <u>http://www.agora-</u>	Feldfunktion geändert
energiewende.de/fileadmin/downloads/publikationen/Agora_Kurzstudie_Entwicklung_der_	
Windenergie_in_Deutschland_web.pdf	
Agora Energiewende (2013b). Kostenoptimaler Ausbau der Erneuerbaren Energien in Deutschland	
http://www.agora-	Feldfunktion geändert
energiewende.de/fileadmin/downloads/presse/Pk Optimierungsstudie/Agora Studie Koste	
noptimaler_Ausbau_der_EE_Web_optimiert.pdf.	
bdew (2012). Anlage zur Presseinformation "Trotz Milliardeninvestitionen keine Entwarnung beim	
Kraftwerksbau" 19. April 2012	
http://www.bdew.de/internet.nsf/id/91C0FC9A8D7AD3EEC12579E9002F8CBF/\$file/120424	Feldfunktion geändert
%20Anlage%20zur%20PM%20Hannover_Kraftwerksliste%20aktuell.pdf	
BMU (2011). Erneuerbare Energien in Zahlen - Nationale und internationale Entwicklung.	
http://www.erneuerbare-	Feldfunktion geändert
energien.de/files/pdfs/allgemein/application/pdf/broschuere ee zahlen bf.pdf	
BMU (2012). Entwicklung der erneuerbaren Energien in Deutschland im Jahr 2011.	
<pre>http://www.bmu.de/files/pdfs/allgemein/application/pdf/ee_in_deutschland_graf_tab.pdf</pre>	Feldfunktion geändert
BMU (2013). Zeitreihen zur Entwicklung der erneuerbaren Energien in Deutschland.	
http://www.erneuerbare-energien.de/unser-	Feldfunktion geändert
service/mediathek/downloads/detailansicht/artikel/zeitreihen-zur-entwicklung-der-	
erneuerbaren-energien-in-deutschland/	

BMWi (2010). Energie in Deutschland. Trends und Hintergründe zur Energieversorgung.	
http://www.bmwi.de/Dateien/Energieportal/PDF/energie-in-	Feldfunktion geändert
deutschland,property=pdf,bereich=bmwi,sprache=de,rwb=true.pdf	
BMWi (2011). Second National Energy Efficiency Action Plan (NEEAP) of the Federal Republic of	
Germany <u>http://www.bfee-</u>	
online.de/bfee/informationsangebote/publikationen/neeaps/zweiter_nationaler_energieeff	
izienz aktionsplan der brd englisch.pdf.	
BMWi (2012). Zahlen und Fakten - Energiedaten - Nationale und Internationale Entwicklung.	
http://www.bmwi.de/BMWi/Navigation/Energie/Statistik-und-	Feldfunktion geändert
Prognosen/Energiedaten/gesamtausgabe.html	
Possard N. (2000) "Capacity factor of wind power realized values vs. estimates." Energy Policy 27:	
2670 2699	
2075-2008.	
Brennan, T. J. (2011). Energy Efficiency Policy: Surveying the Puzzles. Resources for the Future	
Discussion Paper No. RFF DP 11-27. <u>http://rff.org/RFF/Documents/RFF-DP-11-27.pdf</u>	
BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft	(- ,
BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft http://www.solarwirtschaft.de/preisindex	Feldfunktion geändert
BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft http://www.solarwirtschaft.de/preisindex Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to 2030 —	Feldfunktion geändert
BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft <u>http://www.solarwirtschaft.de/preisindex</u> Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to2030 — Update 2009.	Feldfunktion geändert
BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft http://www.solarwirtschaft.de/preisindex Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to 2030 — Update 2009. http://ec.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf	Feldfunktion geändert
BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft http://www.solarwirtschaft.de/preisindex Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to 2030 — Update 2009 . http://cc.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf	Feldfunktion geändert
BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft http://www.solarwirtschaft.de/preisindex Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to 2030 — update 2009 . http://ec.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf dena (2010). dena Grid Study II – Integration of Renewable Energy Sources in the German Power	Feldfunktion geändert
 BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft http://www.solarwirtschaft.de/preisindex Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to 2030 – Update 2009. http://ec.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf dena (2010). dena Grid Study II – Integration of Renewable Energy Sources in the German Power Supply System from 2015 – 2020 with an Outlook to 2025. 	Feldfunktion geändert
BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft <u>http://www.solarwirtschaft.de/preisindex</u> Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to 2030 — Update 2009. <u>http://ec.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf</u> dena (2010). dena Grid Study II – Integration of Renewable Energy Sources in the German Power Supply System from 2015 – 2020 with an Outlook to 2025. <u>http://www.dena.de/fileadmin/user_upload/Publikationen/Sonstiges/Dokumente/Summary</u>	Feldfunktion geändert
BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft http://www.solarwirtschaft.de/preisindex Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to 2030 — Update 2009. http://ec.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf dena (2010). dena Grid Study II – Integration of Renewable Energy Sources in the German Power Supply System from 2015 – 2020 with an Outlook to 2025. http://www.dena.de/fileadmin/user_upload/Publikationen/Sonstiges/Dokumente/Summary_dena_Grid_Study_II.pdf	Feldfunktion geändert
 BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft http://www.solarwirtschaft.de/preisindex Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to2030 – Update 2009. http://ec.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf dena (2010). dena Grid Study II – Integration of Renewable Energy Sources in the German Power Supply System from 2015 – 2020 with an Outlook to 2025. http://www.dena.de/fileadmin/user_upload/Publikationen/Sonstiges/Dokumente/Summary dena_Grid_Study_II.pdf Deutsche Wind Guard (2012). Status des Windenergieausbaus in Deutschland 1. Juni 2012. 	Feldfunktion geändert
 BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft <u>http://www.solarwirtschaft.de/preisindex</u> Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to2030 – Update 2009. <u>http://ec.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf</u> dena (2010). dena Grid Study II – Integration of Renewable Energy Sources in the German Power Supply System from 2015 – 2020 with an Outlook to 2025. <u>http://www.dena.de/fileadmin/user_upload/Publikationen/Sonstiges/Dokumente/Summary_dena_Grid_Study_II.pdf</u> Deutsche Wind Guard (2012). Status des Windenergieausbaus in Deutschland 1. Juni 2012. <u>http://www.wind-energie.de/sites/default/files/attachments/press-release/2012/albers-</u> 	Feldfunktion geändert
 BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft <u>http://www.solarwirtschaft.de/preisindex</u> Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to2030 — Update 2009. <u>http://ec.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf</u> dena (2010). dena Grid Study II – Integration of Renewable Energy Sources in the German Power Supply System from 2015 – 2020 with an Outlook to 2025. <u>http://www.dena.de/fileadmin/user_upload/Publikationen/Sonstiges/Dokumente/Summarydena_Grid_Study_II.pdf </u> Deutsche Wind Guard (2012). Status des Windenergieausbaus in Deutschland 1. Juni 2012. <u>http://www.wind-energie.de/sites/default/files/attachments/press-release/2012/albers-bundesregierung-will-bei-energiewende-keinen-foerderalismus/fact-sheet-status-</u>	Feldfunktion geändert
 BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft <u>http://www.solarwirtschaft.de/preisindex</u> Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to2030 — Update 2009. <u>http://ec.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf</u> dena (2010). dena Grid Study II – Integration of Renewable Energy Sources in the German Power Supply System from 2015 – 2020 with an Outlook to 2025. <u>http://www.dena.de/fileadmin/user_upload/Publikationen/Sonstiges/Dokumente/Summary_dena_Grid_Study_II.pdf</u> Deutsche Wind Guard (2012). Status des Windenergieausbaus in Deutschland 1. Juni 2012. <u>http://www.wind-energie.de/sites/default/files/attachments/press-release/2012/albers-bundesregierung-will-bei-energiewende-keinen-foerderalismus/fact-sheet-status-windenergieausbau-2012-06-30.pdf</u> 	Feldfunktion geändert
BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft http://www.solarwirtschaft.de/preisindex Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to2030 — Update 2009. http://ec.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf dena (2010). dena Grid Study II – Integration of Renewable Energy Sources in the German Power Supply System from 2015 – 2020 with an Outlook to 2025. http://www.dena.de/fileadmin/user_upload/Publikationen/Sonstiges/Dokumente/Summary _dena_Grid_Study_II.pdf Deutsche Wind Guard (2012). Status des Windenergieausbaus in Deutschland 1. Juni 2012. http://www.wind-energie.de/sites/default/files/attachments/press-release/2012/albers- bundesregierung-will-bei-energiewende-keinen-foerderalismus/fact-sheet-status- windenergieausbau-2012-06-30.pdf	Feldfunktion geändert
BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft http://www.solarwirtschaft.de/preisindex Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to2030 — Update 2009. http://ec.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf dena (2010). dena Grid Study II – Integration of Renewable Energy Sources in the German Power Supply System from 2015 – 2020 with an Outlook to 2025. http://www.dena.de/fileadmin/user_upload/Publikationen/Sonstiges/Dokumente/Summarydena_Grid_Study_II.pdf Deutsche Wind Guard (2012). Status des Windenergieausbaus in Deutschland 1. Juni 2012. http://www.wind-energie.de/sites/default/files/attachments/press-release/2012/albersbundesregierung-will-bei-energiewende-keinen-foerderalismus/fact-sheet-statuswindenergieausbau-2012-06-30.pdf Dinica, V. (2011). "Renewable electricity production costs -A framework to assist policy-makers'	Feldfunktion geändert
 BSW Solar (2012). Preisindex Photovoltaik des Bundesverbandes Solarwirtschaft http://www.solarwirtschaft.de/preisindex Capros, P., L. Mantzos, N. Tasios, A. De Vita and N. Kouvaritakis (2010). EU Energy Trends to2030 – Update 2009. http://ec.europa.eu/clima/policies/package/docs/trends to 2030 update 2009 en.pdf dena (2010). dena Grid Study II – Integration of Renewable Energy Sources in the German Power Supply System from 2015 – 2020 with an Outlook to 2025. http://www.dena.de/fileadmin/user_upload/Publikationen/Sonstiges/Dokumente/SummarydenaGrid_Study_II.pdf Deutsche Wind Guard (2012). Status des Windenergieausbaus in Deutschland 1. Juni 2012. http://www.wind-energie.de/sites/default/files/attachments/press-release/2012/albersbundesregierung-will-bei-energiewende-keinen-foerderalismus/fact-sheet-statuswindenergieausbau-2012-06-30.pdf Dinica, V. (2011). "Renewable electricity production costs -A framework to assist policy-makers' decisions on price support." <i>Energy Policy</i> 39(7): 4153-4167. 	Feldfunktion geändert

DLR/IWES/IFNE (2010). Langfristszenarien und Strategien für den Ausbau der erneuerbaren Energien	
in Deutschland bei Berücksichtigung der Entwicklung in Europa und global 'Leitstudie 2010'.	
Projekt Nr. FKZ 03MAP146 des Bundesministerium für Umwelt, Naturschutz und	
Reaktorsicherheit (BMU).	
http://www.bmu.de/erneuerbare_energien/downloads/doc/47034.php	Feldfunktion geändert
DLR/IWES/IFNE (2012). Langfristszenarien und Strategien für den Ausbau der erneuerbaren Energien	
in Deutschland bei Berücksichtigung der Entwicklung in Europa und global. Schlussbericht	
Projekt Nr. FKZ 03MAP146 des Bundesministerium für Umwelt, Naturschutz und	
Reaktorsicherheit (BMU).	
http://www.bmu.de/erneuerbare_energien/downloads/doc/47034.php	Feldfunktion geändert
Edenhofer, O., L. Hirth, B. Knopf, M. Pahle, S. Schlömer, E. Schmid and F. Ueckerdt (under revision).	
"On the Economics of Renewable Energy Sources." <i>Energy Economics</i> .	
EEA (2012). National Renewable Energy Action Plan (NREAP) data from Member States.	
http://www.eea.europa.eu/data-and-maps/figures/national-renewable-energy-action-plan.	
ENTSO-E (2010). Ten Year Network Development Plan 2010-2020. https:// <u>www.entsoe.eu/major-</u>	
projects/ten-year-network-development-plan/tyndp-2012/	
ETG Task Force Energiespeicher (2008). Energiespeicher in Stromversorgungssystemen mit hohem	
Anteil erneuerbarer Energieträger. Bedeutung, Stand der Technik, Handlungsbedarf. VDE	
Studie.	
http://www.vde.com/de/fg/ETG/Arbeitsgebiete/V1/Aktuelles/Oeffentlich/Seiten/Studie-	Feldfunktion geändert
Energiespeicher.aspx	
European Commission (2013a). GREEN PAPER A 2030 framework for climate and energy policies.	
http://eur-lex.europa.eu/LexUriServ/LexUriServ.do?uri=CELEX:52013DC0169:EN:NOT	
European Commission (2013b). Renewable energy progress report.	
http://ec.europa.eu/energy/renewables/reports/doc/com_2013_0175_res_en.pdf	
European Parliament and the European Council (2009). DIRECTIVE 2006/32/EC, Official Journal of	
the European Union. L 114/64.	

FWI/GWS/Prognos (2010) Energieszenarien für ein Energiekonzent der Bundesrenublik	
Deutschland. Projekt Nr. 10/12 des Bundesministeriums für Wirtschaft und Technik (BMWi).	
http://www.bmwi.de/BMWi/Navigation/Service/ publikationen,did=356294.html.	Feldfunktion geändert
Federal Government (2010). Energy Concept for an Environmentally Sound, Reliable and Affordable	
Energy Supply http://www.bmu.de/fileadmin/bmu-	
import/files/english/pdf/application/pdf/energiekonzept bundesregierung en.pdf	
Federal Network Agency (2012). Bericht zum Zustand der leitungsgebundenen Energieversorgung im Winter 2011/2012.	
http://www.bundesnetzagentur.de/SharedDocs/Downloads/DE/BNetzA/Presse/Berichte/20	Feldfunktion geändert
12/NetzBericht ZustandWinter11 12pdf.pdf? blob=publicationFile	
Fink, M. and J. Burck (2009). Energy scenarios in Germany and France - Comparing apples to	
oranges A comparison of the main outcomes. http://www.rac-f.org/enci-	
lowcarb/pdf/04/german_french_energy_scenarios_2009.pdf	
Fischedick, M., R. Schaeffer, A. Adedoyin, M. Akai, T. Bruckner, L. Clarke, V. Krey, I. Savolainen, S.	
Teske, D. Ürge-Vorsatz and R. Wright (2011). Mitigation Potential and Costs. In: IPCC Special	
Report on Renewable Energy Sources and Climate Change Mitigation. O. Edenhofer, R. Pichs-	
Madruga, Y. Sokona, K. Seyboth, P. Matschoss, S. Kadner, T. Zwickel, P. Eickemeier, G.	
Hansen, S. Schlömer and C. von Stechow, Eds. Cambridge, Cambridge University Press,	
United Kingdom and New York, NY, USA.	
Geden, O. and B. Knopf (2013). Die Brüsseler Gefahr.	
http://www.taz.de/1/archiv/archiv/?dig=2013/03/28/a0130. taz. die tageszeitung. Berlin.	Feldfunktion geändert
28.03.2013.	
Gillingham, K., R. G. Newell and K. Palmer (2009). Energy Efficiency Economics and Policy. Resources	
for the Future Discussion Paper No. RFF DP 09-13. <u>http://rff.org/rff/documents/RFF-DP-09-</u>	
<u>13.pdf</u>	
Grubh M. L Edmonde, D. ton Brink and M. Marrison (1992). "The Casts of Limiting Fassil First CO2	
Emissions: A Suprov and Applysis " Applied Poviou of Environment and Possuress 19(1): 207	
178 http://dx.doi.org/10.1146/appurey.eg.18.110193.002145	
-7.0. <u>http://ux.uoi.org/10.1140/annurev.eg.10.110175.002145</u>	
Heptonstall, P., R. Gross, P. Greenacre and T. Cockerill (2012). "The cost of offshore wind:	
Understanding the past and projecting the future." <i>Energy Policy</i> 41 : 815–821.	
30	

Hourcade, JC. and J. Robinson (1996). "Mitigating factors: Assessing the costs of reducing GHG	
emissions. Energy Poincy 24(10-11): 863-873.	
http://www.sciencedirect.com/science/article/pii/s0s01421550000/17	
IEA (2013). Energy Policies of IEA Countries - Germany 2013 Review.	
IWES (2012). Windenergie Report Deutschland 2011.	
http://windmonitor.iwes.fraunhofer.de/bilder/upload/Windreport_2011_de.pdf	Feldfunktion geändert
Junginger, M., P. Lako, S. Lensink, W. van Sark and M. Weiss (2008). Technological learning in the	
energy sector. Report 500102 017. <u>http://www.ecn.nl/docs/library/report/2008/e08034.pdf</u>	
Junginger, M., W. van Sark and A. Faaij, Eds. (2010). Technological Learning In The Energy Sector.	
Lessons for Policy, Industry and Science. Cheltenham, Edward Elgar.	
Knutti, R. (2010). "The end of model democracy?" <i>Climatic Change</i> 102 : 395–404.	
Kofler Energies Power AG (2011). Pressemittleilung: Kofler Energies zieht sich aus dem	
Privatkundengeschäft zurück.	
http://www.koflerenergies.com/pdf/medienmitteilungen/110211 Medienmitteilung Kofler	Feldfunktion geändert
EnergiesAG.pdf	
Masini, A. and F. Menichetti (2012). "The impact of behavioural factors in the renewable energy	
investment decision making process: Conceptual framework and empirical findings." <i>Energy</i>	
Policy 40 : 28-38.	
Neij, L. (2008). "Cost development of future technologies for power generation—A study based on	
2211.	
Nemet, G. F. (2009). "Interim monitoring of cost dynamics for publicly supported energy	
technologies." <i>Energy Policy</i> 37 (3): 825-835.	
Nitsch, J., W. Krewitt, M. Nast, P. Viebahn, S. Gärtner, M. Pehnt, G. Reinhardt, R. Schmidt, A. Uihlein,	
C. Barthel, M. Fischedick and F. Merten (2004). Okologisch optimierter Ausbau der Nutzung	
erneuerbarer Energien in Deutschland.	
Panzer, C. (2012). Future investment costs of renewable energy technologies at volatile energy- and	
raw material prices - an econometric assessment. IAEE Conference, Perth.	
31	

http://www.academia.edu/2110010/The 35 th Annual IAEE International Conference 24 -27 June 2012 Perth Western Australia Perth Convention and Exhibition Centre

- Prässler, T. and J. Schaechtele (2012). "Comparison of the financial attractiveness among prospective offshore wind parks in selected European countries." *Energy Policy* **45**: 86–101.
- Richter, M. (2009). Offshore-Windenergie in Deutschland: Potenziale, Anforderungen und Hürden der Projektfinanzierung von Offshore-Windparks in der deutschen Nord- und Ostsee.
- Schiermeier, Q. (2013). "Renewable power: Germany's energy gamble." *nature* **496**(11 April 2013): 156-158.
- Schlomann, B. and W. Eichhammer (2012). Energy Efficiency Policies and Measures in Germany <u>http://www.isi.fraunhofer.de/isi-media/docs/x/de/publikationen/National-</u> <u>Report Germany November-2012.pdf</u>
- Schmid, E. and B. Knopf (2012). "Ambitious mitigation scenarios for Germany: A participatory approach." *Energy Policy* **51**(0): 662-672. http://www.sciencedirect.com/science/article/pii/S0301421512007653
- Schmid, E., B. Knopf and N. Bauer (2012). REMIND-D: A Hybrid Energy-Economy Model of Germany. Fondazione Eni Enrico Mattei (FEEM) Working Paper Series No. 9.2012. <u>http://www.feem.it/userfiles/attach/20122211032394NDL2012-009.pdf</u>
- Scholz, Y. (2010). Möglichkeiten und Grenzen der Integration verschiedener regenerativer Energiequellen zu einer 100% regenerativen Stromversorgung der Bundesrepublik Deutschland bis zum Jahr 2050. http://www.umweltrat.de/SharedDocs/Downloads/DE/03 Materialien/2010 MAT42 DZLR Integration Energiequellen 2050.pdf? blob=publicationFile
- Sims, R., P. Mercado, W. Krewitt, G. Bhuyan, D. Flynn, H. Holttinen, G. Jannuzzi, S. Khennas, Y. Liu, M. O'Malley, L. J. Nilsson, J. Ogden, K. Ogimoto, H. Outhred, Ø. Ulleberg and F. van Hulle (2011). Integration of Renewable Energy into Present and Future Energy Systems. In: *IPCC Special Report on Renewable Energy Sources and Climate Change Mitigation*. O. Edenhofer, R. Pichs-Madruga, Y. Sokona, K. Seyboth, P. Matschoss, S. Kadner, T. Zwickel, P. Eickemeier, G. Hansen, S. Schlömer and C. von Stechow, Eds. Cambridge, Cambridge University Press, United Kingdom and New York, NY, USA.

Feldfunktion geändert

Sorrell, S. (2007). The Rebound Effect: an assessment of the evidence for economy-wide energy	
savings from improved energy efficiency. UK Energy Research Centre.	
http://www.ukerc.ac.uk/Downloads/PDF/07/0710ReboundEffect/0710ReboundEffectReport	
.pdf	
Sorrell, S., J. Dimitropoulos and M. Sommerville (2009). "Empirical estimates of the direct rebound	
effect: A review." Energy Policy 37 (4): 1356-1371.	
Sorrell, S., E. O'Malley and J. Schleich (2004). The Economics of Energy Efficiency - Barriers to Cost-	
Effective Investment Cheltenham, Edward Elgar.	
SRU (2011). Wege zur 100% erneuerbaren Stromversorgung. Sondergutachten.	
http://www.umweltrat.de/SharedDocs/Downloads/DE/02_Sondergutachten/2011_Sonderg	Feldfunktion geändert
utachten 100Prozent Erneuerbare HD.pdf? blob=publicationFile	
Thollander, P., J. Palm and P. Rohdin (2010). Categorizing Barriers to Energy Efficiency: An	
Interdisciplinary Perspective. In: Energy Efficiency. J. Palm, Ed., SCIYO Books.	
trend:research (2011). Marktakteure Erneuerbare – Energien – Anlagen in der Stromerzeugung.	
www.kni.de/media/pdf/Marktakteure%20Erneuerbare%20Energie%20Anlagen%20in%20der	Feldfunktion geändert
<u>%20Stromerzeugung_03_11_2011.pdf</u>	
Ueckerdt, F., R. Brecha, G. Luderer, P. Sullivan, E. Schmid, N. Bauer and D. Böttger (2011). Variable	
Renewable Energy in modeling climate change mitigation scenarios. International Energy	
Workshop Stanford, USA. <u>http://emf.stanford.edu/files/docs/322/Ueckerdt-Paper.pdf</u>	
van der Zwaan, B., R. Rivera-Tinoco, S. Lensink and P. van den Oosterkamp (2011). Evolving	
Economics of Offshore Wind Power: Cost Reductions from Scaling and Learning.	
http://emf.stanford.edu/files/docs/273/IEW-2011-vanderzwaan-etal.pdf.	
WWF (2009). Modell Deutschland Klimaschutz bis 2050: Vom Ziel her denken.	
<u>http://www.wwf.de/fileadmin/fm-</u>	Feldfunktion geändert
wwf/pdf_neu/WWF_Modell_Deutschland_Endbericht.pdf	
Zeelenberg, S. and J. van der Kloet (2007). Challenging offshore wind: guiding experiences from the	
North Sea region. <u>http://www.offshore-</u>	
power.net/Files/Dok/challengingoffshorewind_complete.pdf	
33	